Author's response to reviews

Title:Eating for 1, Healthy and Active for 2; feasibility of delivering novel, compact training for midwives to build knowledge and confidence in giving nutrition, physical activity and weight management advice during pregnancy.

Authors:

Andrea J Basu (partridgeandrea@hotmail.com)
Lynne Kennedy Prof (l.kennedy@chester.ac.uk)
Karen Tocque Prof (k.tocque@chester.ac.uk)
Sharn Jones (sharn.jones@wales.nhs.uk)

Version:5 Date:16 June 2014

Author's response to reviews: see over
Dear Editorial Team

Re: resubmission of the following research paper to BMC Pregnancy & Childbirth:

Eating for 1, Healthy and Active for 2; feasibility of delivering novel, compact training for midwives to build knowledge and confidence in giving nutrition, physical activity and weight management advice during pregnancy.

(MS: 1085200182111018)

With respect to the comments raised by the editorial panel:

(1) The authors contact details have been amended and the author Basu has cited just one single email address for consistency.

(2) The figure number and title for both figures (1 and 2) have been included within the text at the end of the main manuscript. Neither of the two figures have a legend, only labels for each of the axis; therefore no additional detail has been included.

We hope the above amendments meet with the journals requirements.

Yours Sincerely

Andrea Basu MSc BSc (Hons) RD RNutr (Public Health).